Author's response to reviews

Title: Implementation of collaborative governance in cross-sector innovation and education networks: evidence from the National Health Service in England

Authors:

Pavel V Ovseiko (pavel.ovseiko@medsci.ox.ac.uk)
Catherine O'Sullivan (catherine.o'sullivan@southernhealth.nhs.uk)
Susan C Powell (s.powell@mmu.ac.uk)
Stephen M Davies (stephen.davies@addenbrookes.nhs.uk)
Alastair M Buchan (alastair.buchan@medsci.ox.ac.uk)

Version:2 Date:19 March 2014

Author's response to reviews:

Further to Flory Mae Calumpita's email of 19th March 2014, we have inserted line numbers.